Author's response to reviews

Title: Observational Measure of Implementation Progress: The Stages of Implementation Completion (SIC)

Authors:

   Patricia Chamberlain (pattic@cr2p.org)
   C. Hendricks Brown (chbrown@med.miami.edu)
   Lisa Saldana (lisas@cr2p.org)

Version: 4 Date: 15 August 2011

Author's response to reviews: see over
August 15, 2011

Dr. Anne Sales
Editorial Team
Implementation Science

Dear Dr. Sales:

I have revised the manuscript 1133266443491019 to reflect the requested change by moving the material from the Conclusion section in to a Discussion section. Additionally, the Conclusion section is substantially shortened. Thank you for your review of this manuscript.

Sincerely,

Patricia Chamberlain, PhD
Center for Research to Practice
12 Shelton McMurphey Blvd.
Eugene, OR 97401
pattic@cr2p.org